Author's response to reviews

Title: On the pathogenesis of penile venous leakage: role of the tunica albuginea

Authors:

Ahmed Shafik (shafik@ahmedshafik.com)
Ismail Shafik (shafik@ahmedshafik.com)
Olfat Shafik (shafik@ahmedshafik.com)
Ali Shafik (shafik@ahmedshafik.com)

Version: 5 Date: 3 July 2007

Author's response to reviews:

Dr. Annabel Phillips
Senior Assistant Editor
BMC-series Journals
BioMed Central

Re: MS: 7991061461286921
Shafik et al: On the pathogenesis of penile venous leakage: role of the tunica albuginea

Dear Dr. Phillips,

Thank you for sending us the report of your Reviewer. We have in the meantime re-revised the manuscript in the light of his proposals. We thank your Reviewers for their effort and the time they invested trying to assist us in adding readability and clarity to our article.

With best wishes,
and sincerely,
Ahmed Shafik MD PhD

Answers to Reviewer:

1 - Tunica albuginea was added before TA in page 3, paragraph 2.
2 - Program used to do statistics was added to page 4, paragraph 4.